Author’s response to reviews

Title: Water extract of Cnidii Rhizoma suppresses RANKL-induced osteoclastogenesis in RAW 264.7 cell by inhibiting NFATc1/c-Fos signaling and prevents ovariectomized bone loss in SD-rat

Authors:

Ka-Yeon Lee (togany@hanmail.net)
Jae-Hyun Kim (wogusdlwl222@hanmail.net)
Eun-Young Kim (turns@hanmail.net)
Mijung Yeom (myeam@khu.ac.kr)
Hyuk-Sang Jung (jhs@khu.ac.kr)
Youngjoo Sohn (youngjoos@khu.ac.kr)

Version: 1 Date: 25 Jul 2018

Author’s response to reviews:

Dear Editor,

BMC Complementary and Alternative Medicine

Thank you for your thoughtful comments.
We are very pleased to resubmit our manuscript titled “Cnidii Rhizoma suppresses RANKL-induced osteoclastogenesis in RAW 264.7 cell by inhibiting NFATc1/c-Fos signaling and prevents ovariectomized bone loss in SD-rat” to be considered for publication in the BMC Complementary and Alternative Medicine

We agree your comment and answer the suggestions as follows.
We appreciate for your kind instructions on our paper.

1. Please confirm whether a voucher specimen of the plant material has been deposited in a publicly available herbarium, and include this information in your manuscript. A deposition number should be included, if available.
   -This is described in the "Preparation of CR extracts" section of the methods.
2. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.
-Consent for publication has been revised to "Not Applicable".

3. We would also like to ask for you to provide more justification for the contributions of EYK, as currently he does not automatically qualify for authorship.
-The contribution of EYK has been revised.

4. ARRIVE guidelines
-Guidelines have been completed and attached.

Thank you in advance for reviewing our manuscript.

Sincerely,

Youngjoo Sohn, K.M.D., Ph.D., Professor